Author's response to reviews

Title: Adenovirus type 7 associated with severe and fatal acute lower respiratory infections in argentinian children

Authors:

- Guadalupe Carballal (gcarballal@cemic.edu.ar)
- Cristina Videla (cvidela@cemic.edu.ar)
- Alicia Misirlian (aliciamisir@arnet.com.ar)
- Paula Requeijo (prequeijo@hotmail.com)
- Maria del Carmen Aguilar (mcaguilar@hotmail.com)

Version: 3 Date: 7 Jun 2002

According to the Editorial comments the following changes have been made:
Table 3; Modify the title as follows:
Fatality rate during hospitalization associated to adenovirus genome type in children with acute respiratory infection of lower tract.
Table 2 (footpage): Add the following comment:
Six patients were not followed because their parents took them from our hospital after hospitalization during the acute stage of the disease, therefore the outcome is not available.